Author's response to reviews

Title: Retinal nerve fiber layer thickness and cognitive ability in older people: the Lothian Birth Cohort 1936 study

Authors:

Augustinus Laude (draugustinuslaude@googlemail.com)
Gerassimos Lascaratos (glascaratos@gmail.com)
Ross D Henderson (rhender4@staffmail.ed.ac.uk)
John M Starr (jstarr@staffmail.ed.ac.uk)
Ian J Deary (iand@staffmail.ed.ac.uk)
Baljean Dhillon (Bal.Dhillon@luht.scot.nhs.uk)

Version: 2 Date: 15 June 2012

Author's response to reviews:

The following sections have been added to the manuscript after the Conclusions:

Competing interests

The authors declare that they have no competing interests.

Authors’ contributions

AL is responsible for overall patient recruitment and management of the project and drafted the manuscript. GL is responsible for patient recruitment and data collection. RDH, JMS, IJD and BD are responsible for the design of the study and the co-ordination of resources. All authors read, critically reviewed and approved the final manuscript.

Acknowledgments

The authors would like to thank all the participants from the LBC1936 and acknowledge the help of Alan J Gow, Stuart Gairns and Marion Brannan for their technical assistance with the patient recruitment and data acquisition.